Author's response to reviews

Title: Arthroscopic guided biopsy and radiofrequency thermoablation of a benign neoplasm of the tibial spines area: a treatment option

Authors:

Carmine Zoccali (carminezoccali@libero.it)
Giuseppe Teori (teori@tiscali.it)
Nicola Salducca (salducca@ifo.it)
Berardino Di Paola (dipaoladino@yahoo.it)
Ezio Adriani (eadriani@libero.it)

Version: 2 Date: 20 April 2011

Author's response to reviews:

Dear Reviewer
1) The abstract was modified as asked;
2) The consent was modified including the patient permission for clinical images
3) The sentence 'The authors declare that they have no competing interests” was inserted
4) The authors’ contribution was specified
We hope to have been able to do every modifications required.
Best Regards
Carmine Zoccali, MD